Author’s response to reviews

Title: Identification of $H_2O_2$ induced oxidative stress associated microRNAs in HLE-B3 cells and their clinical relevance to the progression of age-related nuclear cataract

Authors:

Song Wang (wangsongfmmu@126.com)
Chenjun Guo (gchengjun@126.com)
Mengsi Yu (dream8724@163.com)
Xiaona Ning (ningxiaona0223@163.com)
Bo Yan (yb5870@163.com)
Jing Zhao (zhaojing@fmmu.edu.cn)
Angang Yang (agyang@fmmu.edu.cn)
Hong Yan (zeratulws@126.com)

Version: 1 Date: 11 Nov 2017

Author’s response to reviews:

Dear editor, thanks for your patience and kindness. I've revised my manuscript as you required. The corresponding author's e-mail address in the manuscript is now the same as the online version. The "materials & methods" subheading has been amended as "methods" and the "conclusions” subheading was added. Details about patient consent has been added and updated.